Author’s response to reviews

Title: Bayesian designs of phase II oncology trials to select maximum effective dose assuming monotonic dose-response relationship

Authors:

Beibei Guo (beibeiguo@lsu.edu)
Yisheng Li (ysli@mdanderson.org)

Version: 2 Date: 15 February 2014

Author’s response to reviews:

Dear Editorial Office,

I have made all the requested format changes to the manuscript. Specifically, I have:

1) included a title page detailing the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author;
2) restructured the abstract into background, methods, results and conclusions sections.
3) renamed the sections of the manuscript into background, methods, results, conclusions, and list of abbreviations used.
4) included sections of competing interests, authors’ contributions, and acknowledgements.

I hope the revised manuscript now meets the format requirements of the journal, and can move forward in the review process.

Thank you very much for your attention.

Sincerely,

Yisheng Li